Author's response to reviews

Title: Successfull smoking cessation with e-cigarettes in smokers with a documented history fo recurring relapses: case report

Authors:

Pasquale Caponnetto (p.caponnetto@unict.it)
Riccardo Polosa (polosa@unict.it)
Cristina Russo (kristina_russo@yahoo.com)
Davide Campagna (davidecampagna83@gmail.com)

Version: 3 Date: 17 March 2011

Author's response to reviews: see over
Dear Sirs,

In relation to the recently “online submission”, please find enclose 1 copies of our Article entitled “SUCCESSFUL SMOKING CESSATION WITH E-CIGARETTES IN SMOKERS WITH A DOCUMENTED HISTORY OF RECURRING RELAPSES: CASE REPORT”.

Smoking cessation programs are useful in helping smokers to quit, but smoking is a very difficult addiction to break and the need for novel and effective approaches to smoking cessation interventions is unquestionable. The e-cigarette is a battery-powered electronic nicotine delivery device (ENDD), which may help smokers to remain abstinent during their quit attempt. Here, we report for the first time objective measures of smoking cessation in three heavy smokers who experimented the e-cigarette.

1) The manuscript has not been published elsewhere.
2) Conflict of interest: none
3) The manuscript has been read and approved by all authors

With kind regards,

Dr Pasquale Caponnetto
Università di Catania
Dip. di Biomedicina Clinica e Molecolare
University of Catania - ITALY
Email p.caponnetto@unict.it